Author’s response to reviews

Title: Evaluating the Feasibility of a Pharmacist-Guided Patient-Driven Intervention to Improve Blood Pressure Control in Patients with CKD

Authors:

Charles Hopley (charles.hopley@ucdenver.edu)

Emily Andrews (emily.s.andrews@ucdenver.edu)

Patrick Klem (patrick.klem@uchealth.org)

Michelle Jonjak (michellemjonjak@gmail.com)

Ann Grothe (ann.grothe@uchealth.org)

Patrick Ten Eyck (patrick-teneyck@uiowa.edu)

Zhiying You (zhiying.you@ucdenver.edu)

Sarah Billups (sarah.billups@ucdenver.edu)

Corey Lyon (corey.lyon@ucdenver.edu)

Korey Kennelty (korey-kennelty@uiowa.edu)

Bradley Dixon (bradley-dixon@uiowa.edu)

Diana Jalal (diana-jalal@uiowa.edu)

Version: 2 Date: 18 Jan 2019

Author’s response to reviews:

Please see the attached Detailed Response to the Reviewers_R2 document.